Author's response to reviews

Title: Toxicology and Teratology of the Active ingredients of Professional Therapy MuscleCare Products during Pregnancy and Lactation

Authors:

Abdulaziz M. S. Alsaad (ab dulaziz.alsaad@mail.utoronto.ca)
Colleen Fox (colleenfox_19@hotmail.com)
Gideon Koren (gkoren@sickkids.ca)

Version: 6
Date: 8 December 2014

Author's response to reviews: see over
I have reviewed the revisions by the authors and the only outstanding issue is the formatting of the reference list:

1) journal names are given in all lower case (although there is a bit of a mix of capitals and not), publisher names are given in all capitals and lower case, italics and not (eg. ref 38, 42 & 96), placement of the publication year is inconsistent (e.g. ref 56 & 57).

We did corrections as suggested:

1) Reference no. 38, 42, 94, 98 are books.
2) References 56 and 57 are journals.
3) References 23, 24, and 31 are reports.